

Author's response to reviews

Title:Community perceptions of rape and child sexual abuse: a qualitative study in rural Tanzania

Authors:

Muzdalifat S Abeid (muzsalim@yahoo.com)
Projectine Muganyizi (promuga@yahoo.com)
Pia Olsson (pia.olsson@kbh.uu.se)
Elisabeth Darj (elisabeth.darj@kbh.uu.se)
Pia Axemo (pia.axemo@kbh.uu.se)

Version:2**Date:**28 November 2013

Author's response to reviews:

P.O.BOX 71736,
Dar es Salaam,
Tanzania
28th November 2013

To: The Editor-in-Chief
BMC Women's Health

Dear Sir/Madam,

On behalf of my co-others, I am submitting a manuscript entitled "Community perceptions of rape and child sexual abuse: a qualitative study in rural Tanzania" for possible publication in your esteemed journal BMC Women's Health.

I have included the title page at the front of the manuscript file as per editorial requirements. I have also included all the tables at the end of the manuscript file. The authors details have not changed.

We look forward to hearing from you at your earliest convenience.

Yours sincerely,

Muzdalifat Abeid,

Corresponding author. Mobile: +255-754-262411 Email: muzsalim@yahoo.com